Author's response to reviews

Title: Struggling to survive for the sake of the unborn baby: a grounded theory model of exposure to intimate partner violence during pregnancy

Authors:

Hafrun Finnbogadottir (hafrun.finnbogadottir@mah.se)
Anna-Karin E Dykes (anna-karin.dykes@med.lu.se)
Chritine Wann-Hansson (christine.wann-hansson@mah.se)

Version: 2 Date: 21 November 2013

Author's response to reviews:

Dear Editor in Chief,

Unfortunately, we missed some points when submitting our manuscript yesterday. We have completed the submitted revision with the following,

1) We have included a title page with all necessary information about the authors at the front of our manuscript.
2) We have inserted the section Competing Interests after the Conclusion section.
3) We have inserted the section Author’s Contributions and placed it after Competing Interests section.

Thank you for making it possible for us to make these changes.

Best regards,

Anna-Karin Dykes
Hafrún Finnbogadóttir
Christine Wann-Hansson